Author's response to reviews

Title: An investigation of the smoking behaviours of parents before, during and after the birth of their children in Taiwan

Authors:

Shu-Fang Shih (annssshih@nhri.org.tw)
Likwang Chen (likwang@nhri.org.tw)
Chi Pang Wen (cwengood@nhri.org.tw)
Wei-Chih Yang (weichih@nhri.org.tw)
Yaw-Tang Shih (stone@tche.org.tw)

Version: 7 Date: 12 February 2008

Author's response to reviews: see over
Dear Dr. Good,

Thank you very much for accepting our manuscript “An investigation of the smoking behaviours of parents before, during and after the birth of their children in Taiwan” (MS: 6380226851564452) for publication. Regarding the numbers in the table titles, we have re-checked them, and we found that tables are numbered in accordance with the order in which they are referenced in the text. We are not sure whether we caused any confusion regarding this because we did not provide a table for sample characteristics. We actually provided such a table previously, but we later on dropped it due to the request from a reviewer. We hope that this version is now ready for publication after our formatting check. Thank you for all your help.

Yours sincerely,

Likwang Chen, Ph.D.

Associate Investigator
Center for Health Policy Research and Development
National Health Research Institutes
Taiwan